Author's response to reviews

Title: A prospective cohort study of dietary patterns of non-western migrants in the Netherlands in relation to risk factors for cardiovascular diseases: HELIUS-Dietary Patterns

Authors:

Louise H Dekker (l.h.dekker@amc.uva.nl)
Marieke B Snijder (m.b.sniider@amc.uva.nl)
Marja H Beukers (marja.beukers@rivm.nl)
Jeanne HM de Vries (jeanne.devries@wur.nl)
Henny AM Brants (henny.brants@rivm.nl)
Evelien J de Boer (evelien.de.boer@rivm.nl)
Rob M van Dam (ephrmvd@nus.edu.sg)
Karien Stronks (k.stronks@amc.uva.nl)
Mary Nicolaou (m.nicolaou@amc.uva.nl)

Version: 5 Date: 4 June 2011

Author's response to reviews:

Dear editor,

In response to the email of the BMC Series Editorial Production Team dated 28-5-2011 we have taken the opportunity to make some small typographical changes in the manuscript MS ID: 168680746454861.

Yours sincerely, on behalf of all authors,
Louise Dekker